Author's response to reviews

Title: Research-policy partnerships - experiences of the Mental Health and Poverty Project in Ghana, South Africa, Uganda and Zambia

Authors:

Tolib N Mirzoev (t.mirzoev@leeds.ac.uk)
Maye A Omar (m.a.omar@leeds.ac.uk)
Andrew T Green (a.t.green@leeds.ac.uk)
Philippa K Bird (pippabird@gmail.com)
Crick Lund (Crick.Lund@uct.ac.za)
Angela Ofori-Atta (angela.oforiatta@yahoo.com)
Victor Doku (victordoku2003@yahoo.co.uk)

Version: 3 Date: 10 September 2012

Author's response to reviews: see over
Manuscript Title: Research-policy partnerships - experiences of the Mental Health and Poverty Project in Ghana, South Africa, Uganda and Zambia

Dear Editors,

We thank Dr Stephen Hanney for the comment on the presentation of our manuscript. As requested, below we provide the point-by-point descriptions of the actions taken / responses to the comment received.

COMMENT:
The authors have addressed my comments thoroughly and I think the article should be published, after some minor revisions are made to improve consistency in the references in the way some of the journal titles are written. For example, in one place Soc Sci Med is used, and elsewhere Social Science and Medicine; in references 45 and 46 the same journal is written in 2 different ways.

ACTIONS TAKEN:
As recommended, we have now double-checked, and ensured, the consistency of presentation of journal titles in the reference list in the latest version of our manuscript.

Best wishes,

Authors of the manuscript